Author’s response to reviews

Title: Divided by a lack of common language? - a qualitative study exploring the use of language by health professionals treating back pain.

Authors:

Karen L Barker (karen.barker@noc.anglox.nhs.uk)
Margaret Reid (margaret.reid@btinternet.com)
Catherine J Minns Lowe (catherine.minnslowe@noc.nhs.uk)

Version: 8 Date: 28 September 2009

Author’s response to reviews:

Affiliation for Karen Barker & Catherine Minns Lowe amended to include BRU